Author’s response to reviews

Title: Imported Cases of Chikungunya Virus in Iran

Authors:

Mohammad Hassan Pouriayevali (M.h.Pouria@gmail.com)
Farshid Rezaei (Dr.rezaei@live.com)
Tahmineh Jalali (jalalitahmineh@gmail.com)
Vahid Baniasadi (baniasadi.vahid@gmail.com)
Mehdi Fazlalipour (mfp.virology@gmail.com)
Ehsan Mostafavi (mostafaviehsan@gmail.com)
Sahar Khakifirouz (sahar_khakifirouz@yahoo.com)
Tahereh Mohammadi (tahereh.mohammadi12@gmail.com)
Zahra Fereydooni (zahra.fereydooni70@gmail.com)
Mahsa Tavakoli (mahsa.tavakolirad@gmail.com)
Sanam Azad-Manjiri (s.a_92@yahoo.com)
Motahareh Hosseini (mt.hosseini1991@gmail.com)
Mahsa Ghalejoogh (mahsaghalejoogh@gmail.com)
Mohammad Mehdi Gouya (mgouya57@gmail.com)
Anna-Bella Failloux (anna-bella.failloux@pasteur.fr)
Mostafa Salehi-Vaziri (Mostafavaziri1985@gmail.com)

Version: 2 Date: 01 Nov 2019

Author’s response to reviews:

Authors’ Responses to Technical Comments

Dear Editor,

Thank you so much for getting back to us with the technical comments. We appreciate your time and helpful comments. We have addressed the remarks, and hope the revised version of the manuscript meets your expectations.
It is to declare that we have permission to include the names of those mentioned in the Acknowledgements.

Technical Comments:
1. Thank you for your statement in Funding. In addition to the current statement can you also please list the names of the funders?
   Authors’ Response: Done (Page 15, line 269).

2. Please confirm in the covering letter that you have permission to include the names of those mentioned in the Acknowledgements.
   Authors’ Response: Done

3. As one of the authors (Anna-Bella Failloux) is a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.
   Authors’ Response: Done (Page 15, line 292-294).

4. Please state in the figure legend whether the map depicted in figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it. If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time. Should you require an alternative source you may wish to try Wikimedia Commons: http://commons.wikimedia.org/wiki/Main_Page
   Authors’ Response: The figure was originally created in this study and this has been mentioned in the figure legend (Page 19, line 438).

5. Thank you for your current statement in the Ethics approval and consent to participate section. Could you please also include links or references to any specific local legislation regarding national surveillance as well as a statement if any administrative consent was required to access the data?
   Authors’ Response: Unfortunately, we could not find and documented reference regarding ethical legislation in our national surveillance.

6. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letters, responses to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only.
   Authors’ Response: Done